MATHEMATICS 300 — FALL 1997 Introduction to Mathematical Reasoning

Instructor: H. J. Sussmann

HANDOUT NO. 3. September 15, 1997

INSTRUCTOR’S OFFICE HOURS

Monday and Thursday, 4th period. Please let the instructor know after class if you are planning to go see him during office hours, so he will not have to stay in his office unless he knows somebody is coming.

A CORRECTION

Please correct a typo in Handout No. 2. In Page 18, where it says “houndout” it should be “handout)” (with a parenthesis at the end).

HOMEWORK ASSIGNMENT NO. 2.

The second homework assignment is due on Thursday, September 25. It consists of:

a. Problem 4 in Page 16 of the book, all the nonstarred parts, b. Problem 8 in Pages 16-17, all the nonstarred parts, c. Problems 4, 5, 6 in Pages 25-26, all the nonstarred parts. plus d. the five problems in Handout No. 2, pages 17-18.